

Adaptation of complex interventions for people with long-term conditions: a scoping review

Jamal Uddin^{1,2,*,}, Vicky L. Joshi³, Valerie Wells¹, Mithila Faruque⁴, Saidur R. Mashreky⁴, Ani Movsisyan^{5,6}, Rhiannon Evans⁷, Graham Moore⁷ and Rod S. Taylor^{1,8}

¹MRC/CSO Social and Public Health Sciences Unit, Institute of Health and Wellbeing, University of Glasgow, Glasgow, UK, ²Physiotherapy and Cardiac Rehabilitation Unit, Department of Cardiac Surgery, Ibrahim Cardiac Hospital and Research Institute, Dhaka, Bangladesh,

³Department of Physiotherapy and Paramedicine, School of Health and Life Sciences, Glasgow Caledonian University, Glasgow, UK,

⁴Department of Noncommunicable Diseases (NCD), Faculty of Public Health, Bangladesh University of Health Sciences, Dhaka, Bangladesh, ⁵Institute for Medical Information Processing, Biometry and Epidemiology (IBE), Chair of Public Health and Health Services Research, Faculty of Medicine, LMU Munich. Elisabeth-Winterhalter-Weg 6, 81377 Munich, Germany,

⁶Pettenkofer School of Public Health, Faculty of Public Health, Elisabeth-Winterhalter-Weg 6, 81377 Munich, Germany,

⁷Centre for Development, Evaluation, Complexity, and Implementation in Public Health Improvement (DECIPHer), DECIPHer, School of Social Sciences, Cardiff University, Cardiff, UK,

⁸Robertson Centre for Biostatistics, MRC/CSO Social and Public Health Sciences Unit, Institute of Health and Wellbeing, University of Glasgow, Glasgow, UK

*Correspondence address. MRC & CSO Social and Public Health Sciences Unit, Clarice Pears Building, University of Glasgow, 90 Byres Rd, Glasgow G12 8TB, Scotland, UK. Tel: +8801824630856, E-mail: 2780885U@student.gla.ac.uk; uddinhj83@gmail.com

Abstract

Adaptation seeks to transfer and implement healthcare interventions developed and evaluated in one context to another. The aim of this scoping review was to understand current approaches to the adaptation of complex interventions for people with long-term conditions (LTCs) and to identify issues for studies performed in low- and middle-income countries (LMICs). Bibliographic databases were searched from 2000 to October 2022. This review involved five stages: (i) definition of the research question(s); (ii) identifying relevant studies; (iii) study selection; (iv) data charting; and (v) data synthesis. Extraction included an assessment of the: rationale for adaptation; stages and levels of adaptation; use of theoretical frameworks, and quality of reporting using a checklist based on the 2021 ADAPT guidance. Twenty-five studies were included from across 21 LTCs and a range of complex interventions. The majority (16 studies) focused on macro (national or international) level interventions. The rationale for adaptation included intervention transfer across geographical settings [high-income country (HIC) to LMIC: six studies, one HIC to another: eight studies, one LMIC to another: two studies], or transfer across socio-economic/racial groups (five studies), or transfer between different health settings within a single country (one study). Overall, studies were judged to be of moderate reporting quality (median score 23, maximum 46), and typically focused on early stages of adaptation (identification and development) with limited outcome evaluation or implementation assessment of the adapted version of the intervention. Improved reporting of the adaptation for complex interventions. Improved reporting of the adaptation for complex intervention. Improved reporting of the adaptation priorities of the LMIC context.

Lay summary

Limited finance and human capacity may reduce access to new treatments for people with long-term conditions. This is especially true in lowand middle-income countries. One solution is to transfer treatments developed in one place for use in other areas. This paper provides a current summary of international research on adapting treatments. We used a checklist to assess study reporting quality, based on published advice. Our findings showed the need for better conduct and reporting of adaptation. Future guidance should consider the specific needs of low- and middle-income countries.

Keywords: adaptation; complex intervention; long-term conditions; -methodological frameworks; low-and-middle-income countries

© The Author(s) 2024. Published by Oxford University Press on behalf of the Society of Behavioral Medicine.

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/ licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact journals.permissions@oup.com

Graphical Abstract

Adaptation of Complex Interventions For People With Long-Term Conditions: A Scoping Review This study provides a contemporary review of the international practice of intervention adaptation. a rapidly growing field with an evolving methodology that seeks to achieve a better fit between an intervention and a new context. Distribution the characteristics of original intervention of ITCs:

- ⊙ In most of the studies (23/25 studies), the original version of the intervention was a randomised controlled trial.
- A wide range of LTCs, including cancer (3 studies, 12%), dementia (3 studies, 12%), chronic pain (2 studies, 8%), hypertension (2 studies, 8%), and HIV/AIDs (2 studies, 8%).
- ⊙ USA (16 studies, 64%) and UK (2 studies , 8%), and single studies were published in Belgium, Brazil, Nigeria, Pakistan, Portugal, Thailand, and Vietnam

Adapted to complex intervention (stage and level of adaptation and Intervention transfer):

- The majority of the studies' adaptation undertook the early stages; Identification 24 studies, development 18, feasibility 8, evaluation 4, and only 6 implemented the adapted intervention.
- In 17 studies, the level of intervention was at the macro level . 5 was meso and micro 12% level was 3.
- Intervention adaptation (aimed to transfer an intervention) fell into four broad categories- High-income country (HIC) to LMIC (6 studies), HIC to another HIC (8 studies), or LMIC to another LMIC (2 studies), within a single county across either different across socio-economic/racial groups (5 studies), different LTCs (3 studies), or different health settings single study.



Research methods employed:

* All studies used a qualitative research approach (e.g., individual interviews and or focus group discussions with study participants) and eight studies also using quantitative methods.

80% of studies reported using a theoretical framework there was very little consistency in the specific framework.

* Overall, studies were judged to be of moderate reporting quality (median score of 23 out of a maximum of 46).

Implications

Practice: This scoping review provides a comprehensive and contemporary overview of the practice of adapting complex interventions for people with long-term conditions.

Policy: Our findings provide a resource for researchers, policymakers, and practitioners adapting healthcare interventions to new contexts, particularly between low-, middle-, and high-income countries.

Research: Future development of adaptation methods guidance requires consideration of the needs and priorities of low- and middle-income countries.

Introduction

The provision of effective evidence-based healthcare services for people with long-term conditions (LTCs) is one of the key priorities facing healthcare systems across the world [1, 2]. LTCs, such as diabetes, heart failure, and chronic pain, often require sustained engagement with the healthcare delivery system and support to enable people to manage their condition(s) [3]. It is estimated that LTCs contribute to 60% of deaths and 46% of the global health burden with much of this impact occurring in low- and middle-income countries (LMICs) [4, 5]. Alongside the growing burden of LTCs, is the challenge of constrained finance and human capacity in the development and evaluation of *de novo* interventions and the provision of healthcare services more broadly. One potential solution to these challenges is intervention adaptation, which seeks to transfer and implement healthcare interventions developed and evaluated in one context to another [6, 7]. Adaptation is a process of modification to the original intervention content and/or its delivery to fit an alternative context or study population/disease group [8, 9]. The use of interventions with a previous evidence base in new contexts might be more efficient than developing new interventions and increase the

chances of maintaining effectiveness and ensuring success in implementation. Given that LMICs face the combined pressures of a growing burden of LTCs and highly constrained finance and human capacity, the adaptation of the existing intervention approach is likely to be especially important and necessary in this setting [10, 11]. Intervention adaptation has been described as a process that involves 'intentional modification(s) of an evidence-informed intervention, to achieve a better fit between an intervention and a new context. Modification can include planned adaptations (changes made before introducing a new intervention) and responsive adaptations (changes made intentionally but in response to emerging contextual issues occurring during implementation)' [7, 12].

The science of adaptation of healthcare interventions is a rapidly growing field with an evolving methodology [9, 12–14]. The ADAPT framework published in the British Medical Journal in 2021 is widely recognized as a key source of consensus-informed guidance for adapting and transferring healthcare interventions to new contexts [15]. ADAPT seeks to provide step-by-step guidance for working with stakeholders, selecting suitable interventions, undertaking adaptations, making decisions on evaluation and implementation, and reporting adapted interventions.

Whilst a small number of previous reviews have assessed the reporting and methodology of studies describing the adaptation of healthcare interventions, none to date have focused on LTCs or considered the implications for the conduct of adaptation studies in LMICs [6, 7].

The aim of this scoping review was to understand current approaches to the adaptation of complex healthcare interventions for people with LTCs. Key research questions were: (i) What is the rationale for adaptation of complex interventions for LTCs? (ii) What research methods are used by adaptation studies? (iii) Do adaptation studies use frameworks? (iv) How well do the conduct and reporting of adaptation studies conform to 2021 ADAPT guidance? In addressing these research questions, we sought to identify the specific issues and challenges to conducting adaptation of complex healthcare interventions in the context of the LMIC setting.

Methods

Study design

To address our study aims and research questions, a scoping review was undertaken [16]. We used a methodological framework as initially proposed by Arksey and O'Malley [16] and adapted by Levac *et al.* [17] and Colquhoun *et al.* [18]. The review comprised five stages: (i) definition of the research question(s); (ii) identification of relevant studies; (iii) study selection; (iv) data charting; and (v). data synthesis. The study is reported in accordance with the Preferred Reporting Items for Systematic Reviews for Scoping Reviews (PRISMA-ScR) extension [19].

Identifying relevant studies

The study inclusion and exclusion criteria were adapted from Movsisyan et al. [20] and are summarized in Supplementary Table 1. A list of eligible LTCs was compiled by combining conditions listed by the Cambridge Multimorbidity Score and Barnett et al. (see Supplementary Table 2) [21, 22]. The following electronic databases were searched: Medline, CINHAL, PsycINFO, and Cochrane Library. The research strategy was designed with an experienced information specialist. The search strategy development followed an iterative piloting process and was modified to ensure that we identified appropriate literature based on a small number of adaptation studies already known to the research team. Details of searches are provided in Supplementary Tables 3a-c. To reflect the recent development of the field of intervention adaptation in healthcare, databases were searched from 1 January 2000 to 3 October 2022. We limited inclusion to studies published in English.

Study selection

Search results were exported into Covidence (Veritas Health Innovation Ltd, Melbourne, Australia) [23], where duplicates were removed. Two reviewers independently undertook study screening based on the inclusion and exclusion criteria; a third reviewer resolved any conflicts.

Data charting

Data were extracted by a single reviewer and checked by a second. Key domain data were extracted in accord with the study research questions. General study characteristics included details of the population and intervention (original and adapted). The level of intervention was assessed as—'micro', i.e., intervening with individuals and their immediate social network and relationships; 'meso', i.e., intervening with medium-level population groups and institutional or cultural change; or 'macro', i.e., operating at the national or global level, such as through regulations, taxation, other government policies, or mass media [24]. In addition, details of the rationale for adaptation, methods of adaptation (use of qualitative/ quantitative research; stages of adaptation process addressed; level of evidence for the original intervention), use of theoretical frameworks, and reporting of the adaptation process. Studies were assessed as to whether they undertook the following five stages of the adaptation process: (i) identification (identifying the factors that would need to be addressed to adapt an intervention); (ii) development (the process of developing an adapted version of the intervention); (iii) feasibility (assessment of the acceptability or feasibility of the adapted intervention); (iv) evaluation (assessment of the efficacy/safety of the adapted intervention); and (v) implementation (a scaled roll out of the adaptation into 'real world' practice) [25].

Reporting of the adaptation process and methodology was assessed using a checklist developed by a group of authors based on the 2021 ADAPT guidance [15]. Whilst published in 2021, we used the checklist developed from the ADAPT guidance to retrospectively assess the quality and transparency of reporting of our included studies. This checklist directly maps to the items included in the ADAPT guidance including details of forming an adaptation team (four items); rationale for the intervention adaptation-context fit (six items); methods of intervention adaptation (five items); methods of intervention evaluation (six items) and plans for implementation and maintenance of the adapted intervention (two items). Each checklist item was judged as 'fully met' (score: 2), 'partially met' (score: 1), or 'not reported' (score: 0). Item scores were totalled with a possible total checklist score ranging from 0 to 46. The reporting checklist template is shown in Supplementary Table 4. For each item, where available, details from each individual publication were extracted as evidence to support the scoring decision. The checklist was piloted across three studies by each of the three reviewers. Following this piloting process, we finalized the wording of the checklist items. No formal quantification of checklist scoring agreement between reviewers (e.g., Kappa score) was performed.

Data extraction was undertaken by a single reviewer using a standardized pre-piloted Excel proforma and checked by a second reviewer. Where there were disagreements between the two reviewers, discussion took place until a consensus was reached.

Data synthesis

Given this is a scoping review, the focus of data presentation and synthesis was a descriptive narrative analysis supported by the presentation of tabular and graphical summaries of included studies that directly address the study research questions. Findings are presented using descriptive statistics, including frequency counts (and percentages) and medians.

Results

Study selection

The results of the search and study selection process are presented in a PRISMA flow diagram (see Fig. 1). Of the 1020 titles and abstracts identified, a total of 25 adaptation studies (30 publications) were included. The two main reasons for exclusion





Figure 1 PRISMA flow diagram reporting study selection

were that studies did not focus on a LTC population, or the purpose of the study was not about intervention adaptation.

General characteristics of included studies

The study characteristics of the included studies are detailed in Table 1 [26–55]. Six studies were published up to 2015 and 19 studies between 2016 and 2022. The two main study countries of publication were the USA (16, 64%) and the UK (2, 8%), with one study from Belgium, Brazil, Nigeria, Pakistan, Portugal, Thailand, or Vietnam. Studies included a wide range of LTCs, including cancer (3, 12%), dementia (3, 12%), chronic pain (2, 8%), hypertension (2, 8%), and HIV/AIDs (2, 8%). A wide range of complex interventions were employed across studies that included the promotion of mental health, disease self-management, and medication adherence.

Rationale for adaptation

The rationale for intervention adaptation fell into four broad categories (see Table 1). Most studies (16, 64%) took an

intervention developed in one country and adapted the intervention for the same LTC population in another country (or group of countries)—high-income country (HIC) to LMIC (six studies), one HIC to another HIC (eight studies), or one LMIC to another LMIC (two studies). The remaining studies aimed to transfer an intervention within a single county across either different across socio-economic/racial groups (five studies), different LTCs (three studies), or different health settings (one study).

Adaptation study methods

Nature of original intervention evidence

For the majority of studies (23 studies), the evidence base for the original version of the intervention was a randomized controlled trial (see Table 1).

Level of adaptation

For 17 studies, the level of intervention was at the macro level (e.g., Carver *et al.* [37], transfer from Iceland to Scotland), five studies were at the meso level (e.g., Cassel *et al.* [28], different

Author (year),	Population	Intervention			Rationale for adaptation	Level of evidence	Intervention
country of publication		Description	Original version name	Adapted version name		intervention	ICVCI-
O'Donnell (2022), UK	Heavy alcohol use and depression	Intervention and training pro- gramme for the treatment and prevention of comorbid heavy drinking and depression	Scale-up of Prevention and Management of Alcohol Use Disorders and Comor- bid Depression in Latin America (SCALA)	Not stated	Different geographic setting (UK to Latin America)	Quasi experimen- tal study	Macro
Atif (2020), Pakistan	Anxiety during pregnancy	Psychosocial intervention based on principles of cognitive behaviour therapy	Thinking Healthy Pro- gramme (THP)	Happy Mother, Healthy Baby	Different geographic setting (HICs to Pakistan)	Systematic review	Macro
Bertrand (2019)/ Marinho (2021), Brazil	Dementia	Intervention to improve cognition and quality of life	Cognitive stimulation ther- apy (CST)	Not stated	Different geographical setting (UK to Brazil)	Randomized con- trolled trial	Macro
Bornheimer (2022), USA	Schizophrenia spectrum and other Psychosis disorders	Intervention based on cognitive behavioural methods to prevent suicide in community mental health settings	Cognitive Behavioural Suicide Prevention for psychosis	Not stated	Different geographical setting (UK to USA)	Randomized controlled trial (pilot)	Macro
Carver (2021), UK	Substance use	Intervention for preventing sub- stance use among young people	The Icelandic Model (IM)	The Icelandic Model in Scotland	Different geographical setting (Iceland to Scotland)	Randomized con- trolled trial	Macro
Cassel (2014), USA	Obesity and cancer	Intervention for reducing obesity among cancer patients	Body and Soul	Not stated	Different racial group (Afri- can Americans to Samoans population in same coun- trv—USA)	Randomized con- trolled trial	Meso
Chen (2012)/ Parker (2012), USA	Arthritis	Arthritis self-management pro- gramme for older adults	Evidence-based interventions of arthritis	Arthritis Self-Help Program for older African Ameri- can, Hispanic and non-Hispanic white adults	Different racial group (Afri- can American, Hispanic and non-Hispanic white adults) in same country— USA)	Randomized con- trolled trial	Meso
Cho (2020), USA	Prostate cancer and partners	Intervention for cancer survivors' and their partners' potential unmet needs including anxiety/ uncertainty about cancer progression, communication between partners, cultural sen- sitivity, and of motivation and behaviours between partners	Active Living After Cancer (ALAC)	Watchful Living	Different disease population (different cancers) in same population country (USA)	Randomized con- trolled trial	Micro

Table 1 Characteristics of included studies and adaptation methods

Author (year),	Population	Intervention			Rationale for adaptation	Level of evidence	Intervention
country of publication		Description	Original version name	Adapted version name		for the original intervention	level"
Fort (2019), USA/ Paniagua-Avila (2020), Guatemala	Hypertension	Programme to control blood pressure	Hypertension Control Program in Argentina by Community Health Worker (CHW)-Led	Not stated	Different geographical setting (Argentina to Guatemala)	Randomized controlled trial	Macro
Gorman (2021), USA	Cancer	Intervention to improve aspects of sexual health, including sexual arousal, desire, satisfaction, and overall sexual functioning	Mindfulness-based interven- tions (MBIs) of different cancer men and women in USA	Not stated	Different disease group (different cancers) in same country—USA	Randomized con- trolled trial	Micro
Greenberg (2019), USA	Heterogeneous Chronic Pain	Intervention programme to integrate physical activity with mind-body skills to increase comprehensive physical and emotional functioning of chronic pain	Mind-body interventions	Get Active and Get Active With Fitbit Programs	Different disease group (generic disease to pain) in same country (USA)	Randomized con- trolled trial	Micro
Alvares Pereira (2022), Portugal	Dementia	Intervention to improve cognitive function and quality of life	Cognitive Stimulation Therapy	Making a Difference	Different geographical setting (UK to Portugal)	Randomized con- trolled trial	Macro
Hopkins (2022), USA	People at risk of obesity, CVD, and diabetes	Intervention to promote a healthy diet and lifestyle	Celebremos La Salud	Not stated	Different population in a same county (Latino to Alaskan Yup'ik population in USA)	Randomized con- trolled trial	Meso
Jans (2020), Belgium	Chronic condi- tions	Intervention to promote self- management in community- setting	Chronic Disease Self- Management Program of Stanford University (CDSMP)	Not stated	Different geographical setting (USA to Belgium)	Randomized con- trolled trial	Macro
Kangovi (2016), USA	Chronic illness	Intervention to help patients track progress toward their chronic disease management goals to motivate health behaviour change	Community health worker (CHW) programs	IMPaCT program	Different settings in same country (hospital to community-setting -USA)	Randomized con- trolled trial	Meso
Magidson (2014), USA	HIV/AIDS with depression	Intervention to address improve- ments in depressive symptoms and adherence	Behavioural activation (BA) and Life-Steps	Act Healthy Group integrated BA and Life-Steps treatment	Different setting (mental health setting to general medical setting in the same country—USA)	Randomized con- trolled trial	Meso
Muroff (2017), USA	Alcohol and other drug/mental disorders	Programme to assist recovery, con- tinuing their access to resources, case management, and quality information after leaving resi- dential treatment	A smartphone application, Addiction–Comprehen- sive Health Enhancement Support System (A-CHESS)	CASA-CHESS	Different racial population (English speaking Hispan- ics to Spanish speaking Latinos) in same country (USA)	Randomized con- trolled trial	Meso

Table 1. Continued

Author (year),	Population	Intervention			Rationale for adaptation	Level of evidence	Intervention
country or publication		Description	Original version name	Adapted version name		ior the original intervention	"level"
Ojo (2020), USA	Stroke prevention	Programme for optimal hyperten- sion management	Discharge Education Strate- gies for Reduction of Vas- cular Events (DESERVE)	Not stated	Different geographical set- tings (USA to Ghana)	Randomized con- trolled trial	Macro
Okoli (2021), Nigeria	Hypertension	Programme to improve hyperten- sion diagnosis, treatment, and control and reduce deaths from cardiovascular diseases	US Kaiser Permanente North- ern California (KPNC) model and (WHO) HEARTS	Not stated	Different geographical set- tings (USA to Nigeria)	Randomized con- trolled trial	Macro
Olson (2022), USA	Type 2 Diabetes	Web-Based Self-management education programme to sup- port the management of Type 2 diabetes	Diabetes Education and Self-Management for Ongoing and Newly Diag- nosed (My DESMOND)	Not stated	Different geographical set- tings (UK to Australia)	Randomized con- trolled trial	Macro
Risendal. (2014, 2015), USA	Cancer	Intervention for chronic disease self-management programme for cancer survivors	Cancer Thriving and Surviv- ing Program (CTS)	Chronic Disease Self-Management Program (CDSMP) for cancer survi- vors' adaptation	Different geographical setting (UK to USA)	Randomized con- trolled trial	Macro and micro
Tongsiri (2022)/ Chen (2022), Thailand	Dementia	Intervention for dementia to reduce behavioural and psycho- logical symptoms of dementia	Reducing Disability in Alz- heimer's Disease (RDAD)	Thai Reducing Disability in Alzheimer's Disease (Thai-RDAD)	Different geographical set- tings (USA to Thailand)	Randomized con- trolled trial	Macro
Tran (2022), Vietnam	HIV and AIDS with depression, anxiety, and/or stress disorder	Programme to improve common mental disorders (CMDs)	Friendship Bench (FB), a problem-solving therapy	Not stated	Different geographical settings (Zimbabwe to Vietnam)	Randomized con- trolled trial	Macro
Wechsberg (2015), USA	High risk of HIV/ AIDS	Behavioural intervention for couples and the men as partners programme to address the interrelated issues of substance use, sex risk behaviours, gender role expecta- tions and <i>conder-based</i> violence	Western Cape Women's Health, program	Not stated	Different geographical set- tings (USA to South Africa)	Randomized con- trolled trial	Macro
Williams (2013), USA	HIV/AIDS self- reporting poor adherence to antiretroviral drugs	Nursing intervention to improve medication adherence	Adherence Through Home Education and Nursing Assessment (ATHENA)	Not stated	Different geographical set- tings (USA to China)	Randomized con- trolled trial	Macro

"Level of intervention—Macro: operating at the national or global level, such as through regulations; Meso: intervening with medium-level population groups, institutional or cultural change; Micro: intervening with individuals and their immediate social network and relationships.

Table 1. Continued

racial groups—African American to Samoans population—in the same country—USA), and three studies were at a micro level [e.g., Gorman *et al.* [45], different disease groups (different cancers) in the same country—USA] (see Table 1).

Stages of adaptation process and research methods

Whilst the majority of included studies undertook the early stages of adaptation, i.e., identification (24, 96%) and development (18, 72%), few studies undertook the later stages of either assessment of feasibility (8, 32%) or evaluation (4, 16%), and only 6 (24%) reported proceeding to implement the adapted intervention (see Table 2). None of the studies stated that they had implemented the adapted intervention or reported testing either the feasibility or effectiveness of the adapted version.

Research methods employed

All studies used a qualitative research approach (e.g., individual interviews and or focus group discussions with study participants) and eight studies also using quantitative methods (e.g., randomized and non-randomized pilot trials) (see Table 2). Whilst it was consistently reported that the original intervention had been tested using quantitative methods (e.g., Randomised Controlled Trials (RCTs) none of the studies provided information that the adapted interventions have been formally tested for feasibility, and/or efficacy/effectiveness.

Use of adaptation frameworks

Although twenty (80%) studies reported using a theoretical framework there was very little consistency in the specific framework that was cited across studies (see Table 3).

Quality of reporting and conformance with ADAPT guidance checklist

Figure 2 shows how adaptation studies conformed with the ADAPT 2021 guidance [25]. The median total ADAPT reporting checklist score across studies was 23 (range 11-32) out of a maximum of 46. The three initial domains of the checklist (i.e., 'forming an intervention team'; 'assessment of the rationale for intervention and context fit'; and 'planning and undertaking the adaptation') were generally well reported whilst the latter two domain items ('planning/ undertaking an evaluation' or 'implementing/maintaining the intervention at scale') more poorly reported. However, some of the specific checklists within the first three domains were consistently poor or not reported across most studies, i.e., item 1C-working with (original intervention) developers and handling conflicts of interest; item 2F-intellectual property issues around the adapted intervention; item 3Cunintended contexts. There was no evidence of a difference in the quality of reporting of studies according to their publication date. For each of the checklist domain items, examples of good ('fully met') reporting were extracted verbatim

Table 2 Stage of adaptation and research methods employed of included studies

Author (year)	Stage of adapta	ation				Research methods
	Identification	Development	Feasibility	Evaluation	Implementation	_
O'Donnell (2022)	J	J	х	Х	x	Qualitative
Atif (2020)	х	J	х	J	х	Qualitative
Bertrand (2019)/Marinho (2021)	J	х	х	х	Х	Qualitative
Bornheimer (2022)	J	х	J	х	J	Mixed
Carver (2021)	J	х	х	х	х	Qualitative
Cassel (2014)	J	х	J	х	х	Qualitative
Chen (2012)/Parker (2012)	J	J	х	х	х	Mixed
Cho (2020)	J	J	х	х	х	Qualitative
Fort (2019)/ Paniagua-Avila (2020)	J	х	х	х	х	Qualitative
Gorman (2021)	J	J	J	х	х	Mixed
Greenberg (2019)	х	J	J	х	Х	Qualitative
Alvares Pereira (2022)	J	J	х	х	х	Qualitative
Hopkins (2022)	J	J	х	х	х	Qualitative
Jans (2020)	J	J	х	х	Х	Qualitative
Kangovi (2016)	J	Х	J	х	J	Mixed
Magidson (2014)	J	J	J	х	х	Mixed
Muroff (2017)	J	J	х	х	J	Mixed
Ojo (2020)	J	Х	J	х	J	Qualitative
Okoli (2021)	J	Х	х	х	J	Qualitative
Olson (2022)	J	J		J	Х	Qualitative
Risendal (2014 and 2015)/Tongsiri (2022) and Chen (2022)	J	J	J	J	Х	Mixed
Tran (2022)	J	J	х	x	J	Qualitative
Wechsberg (2015)	J	J	x	х	х	Qualitative
Williams (2013)	J	J	x	х	х	Qualitative

J: adaptation stage reported; x: adaptation stage not reported.

Tab	le 3	Reference	to a	adaptation	framework	of	included	studies
-----	------	-----------	------	------------	-----------	----	----------	---------

Author (year)	Use of theoretical framework	Frameworks used
O'Donnell (2022)	J	Community-based developmental approach to adapt CST to different cultures and the Formative Method for Adapting Psychotherapy (FMAP)
Atif (2020)	J	Medical research Council MRC (UK) framework for development and evaluation of complex interventions
Bertrand (2019)/Marinho (2021)	х	
Bornheimer (2022)	J	Community-based participatory research (CBPR) methods
Carver (2021)	J	Consolidated Framework for Implementation Research (CFIR)
Cassel (2014)	J	Community-based participatory research (CBPR)
Chen (2012)/Parker (2012)	J	The Method for Program Adaptation through Community Engagement (M-PACE)
Cho (2020)	J	Intervention Mapping Adapt (IM Adapt) and Typology of Adaptation
Fort (2019)/Paniagua-Avila <i>et al.</i> (2020)	J	RE-AIM (reach, effectiveness, adoption, implementation, maintenance) framework
Gorman (2021)	J	ADAPT-ITT model and Modifications were tracked and coded according to the Frame- work for Reporting Adaptations and Modifications expanded (FRAME)
Greenberg (2019)	J	National Institute of Health (NIH) stage model for behavioural intervention develop- ment, National Institute of Complementary and Integrative Health (NCCIH) model for developing and testing mind-body intervention
Alvares Pereira (2022)	J	Barrera and Castro framework
Hopkins (2022)	J	Modified versions of Adapted Intervention Mapping (IM) approach
Jans (2020)	J	Goldstein Framework
Kangovi (2016)	х	
Magidson (2014)	х	
Muroff (2017)	J	The cultural adaptation stage model
Ojo (2020)	J	FRAME framework for reporting adaptations to evidence-based interventions
Okoli (2021)	J	Primary Health Care Performance Initiative (PHCPI) conceptual framework and mapped onto Consolidated Framework for Implementation Research (CFIR) main domains
Olson (2022)	х	
Risendal (2014, 2015)	х	
Tongsiri (2022)/Chen (2022)	J	Different theoretical frameworks
Tran (2022)	J	Assessment-Decision-Adaptation-Production-Topical (ADAPT-ITT) framework
Wechsberg (2015)	J	ADAPT Framework
Williams (2013)	J	Castro and Barerrera framework

J: framework reported; x: framework not reported.

from the included studies and are provided in Supplementary Table 5.

Discussion

This scoping review provides a contemporary synthesis of 25 international studies published since 2000 and reports on the adaptation of complex interventions for people with LTCs. Our review included adaptation studies across a wide range of interventions and diseases, typically focusing on a macro (or national) level. The most common rationale for adaptation was transferring an intervention from one geographical setting (e.g., from an HIC to an LMIC) or across ethnic or LTC groups. Although the majority of studies referred to an underpinning theoretical framework, there was no consistency in reporting. Whilst studies addressed all five stages of the adaptation process (i.e., identification, development, feasibility, evaluation, and implementation) [24] we found the focus was often only on the first three stages. Using a checklist developed from the ADAPT 2021 guidance, the overall

quality of study reporting was judged to be moderate, and a number of reporting items were consistently omitted. A key strength of the studies identified by our review was the large proportion where the adaptation was based on an original intervention with evidence of efficacy/effectiveness assessed based on a randomized trial design.

Comparison to current knowledge

Our findings have some similarities and differences with the limited number of previous scoping reviews of adaptation studies published to date [7, 12, 20]. Movsisyan reported that 12 of their 28 (43%) included studies described the transfer of a public health intervention from one country to another, the remainder examining adaptations across different population groups within the same country. In contrast to this study, both the ADAPT 2021 guidance publication [12] and Movsisyan [20] found macro (or national)-level interventions to be relatively rare. Similarly, the ADAPT 2021 guidance group noted the reporting of several adaptation frameworks but there was no consensus. Our finding that the majority of

First author (year)	Fo ad tea	ormi lapta am	ng a atio	n n	A ra in co in	sse itior iterv onsi iterv	ssm nale /ent ider /ent	for ion, atio	of th the and n of cont	the ext fit	P ui ai	lanr ndei dapt	ing rtaki atio	and ing f n	the	Pla un ev	anni Ideri alua	ing a taki atior	and ng t า	he		Impleme and maintain the interver	enting ning ntion	Total checklist score
	1 A	1 B	1 C	1D	2 A	2 B	2 C	2 D	2 E	2F	3 A	3 B	3 C	3 D	3E	4 A	4 B	4 C	4 D	4 E	4 F	5A	5B	
Chen (2012)/ Parker (2012)	2	2	2	2	2	0	2	2	1	0	2	2	0	2	1	0	0	0	0	0	0	0	0	22
Williams (2013)	2	2	2	0	2	2	2	2	2	0	2	0	0	1	1	2	1	0	0	0	1	0	0	24
Magidson (2014)	2	1	0	0	2	1	1	2	2	0	1	2	0	0	0	0	1	1	0	0	0	0	0	16
Cassel (2014)	2	2	0	1	2	0	1	2	2	0	1	0	0	1	2	2	1	0	0	0	1	0	0	20
Risendal (2014, 2015)	2	1	1	1	2	1	1	2	2	0	0	2	0	1	2	1	2	0	0	0	0	0	0	21
Wechsberg (2015)	2	2	2	2	2	2	2	2	1	0	2	2	0	0	2	0	1	0	2	0	0	1	0	27
Kangovi (2016)	1	0	0	0	2	0	2	2	0	0	1	1	0	1	1	0	0	0	0	0	0	0	0	11
Muroff (2017)	2	2	0	2	2	2	0	0	2	0	2	2	0	1	2	1	0	0	1	1	0	0	0	22
Greenberg (2019)	1	1	0	0	2	1	1	1	0	0	1	2	0	0	1	2	1	0	0	0	1	0	0	15
Fort (2019) / Paniagua (2020)	2	2	2	2	2	0	1	2	2	0	2	2	0	0	0	2	2	2	0	0	0	1	2	28
Bertrand (2019)/Marinho (2021)	1	2	0	1	2	1	2	2	2	0	2	2	0	0	1	2	2	0	1	0	1	0	0	24
Atif (2020)	2	2	0	0	2	2	1	1	2	0	1	1	0	1	1	2	1	1	1	0	2	0	0	23
Cho (2020)	2	1	2	2	2	2	1	2	2	0	2	2	0	1	2	2	0	0	0	0	0	1	0	26
Jans (2020)	2	1	0	1	1	1	1	2	2	0	2	2	0	1	2	2	1	0	2	0	0	1	0	24
Ojo (2020)	2	0	0	0	2	1	2	2	2	0	2	1	0	2	2	0	1	0	0	0	0	0	0	19
Carver (2021)	1	0	0	0	2	1	2	2	2	0	2	2	0	0	1	2	2	0	0	0	0	2	2	23
Gorman (2021)	2	0	0	1	2	1	1	1	2	0	2	2	0	0	0	1	1	0	0	0	0	0	0	16
Okoli (2021)	2	1	0	0	2	1	2	2	1	0	1	1	0	1	2	2	2	0	0	0	2	0	0	22
Olson (2022)	2	2	2	1	2	1	2	1	2	0	2	2	2	1	1	2	2	0	2	0	0	2	0	31
O'Donnell (2022)	2	2	0	1	2	2	2	2	2	0	2	2	0	1	1	0	1	1	0	0	1	0	0	24
Bornheimer (2022)	2	2	2	2	2	2	2	2	2	0	1	1	0	1	1	0	0	1	2	0	1	0	0	26
Alvares Pereira (2022)	2	2	0	0	2	2	1	1	2	0	2	2	0	1	1	1	1	0	2	0	0	1	0	23
Hopkins (2022)	2	0	0	1	1	2	1	0	0	0	1	2	0	0	1	0	0	0	0	0	0	0	0	11
Tongsiri (2022)/Chen H (2022)	2	2	2	1	2	2	1	2	2	0	1	2	0	1	2	2	2	0	1	0	2	1	0	30
Tran (2022)	2	2	2	1	2	2	2	2	2	0	2	2	0	1	2	2	2	0	2	0	2	0	0	32
Number of studies that achieve the highest score 2 (fully reported) of each of the domain items of the ADAPT checklists	2 1	1 4	9	6	2 3	1 1	1 2	1 7	1 9	0	1 5	1 8	1	2	10	1 3	9	1	6	0	4	2	2	Median: 23 Range of all 25 studies: 11-32

*Score 2: Fully reported, score 1 - Partially reported, and score 0 - Not reported

Figure 2 A summary of ADAPT checklist assessment of the quality of reporting of included studies. *Score 2: fully reported, score 1—partially reported, and score 0—not reported

included studies focused on the early stages of adaptation of identification and development (24, 96%) is consistent with Movsisyan [20], and the review of adaptation frameworks by Escoffery [7]. Given the 2021 publication date of the ADAPT guidance and that our study included publications from 2000 to 2022, it is perhaps not surprising that none of them directly referenced this guidance. However, this may simply reflect the fact that the guidance was only published in the last 2–3 years and its uptake will be seen in future adaptation studies.

Strengths and limitations

This scoping review has several strengths. It provides a comprehensive and contemporary overview of the international literature on complex intervention adaptation studies for people with LTCs. Second, this study addresses some gaps in the use of theoretical and methodological guidance for intervention adaptation for which there is no current consensus on best practice. Third, we have developed and applied a checklist for describing the quality of reporting based on the ADAPT 2021 guidance [25]. Given the likely growing importance and utilization of intervention adaptation approaches [8, 25, 56], there is a need for the adoption of rigorous methodology approaches across the research community. Our checklist developed based on the ADAPT 2021 guidance provides a potential tool to help assess and quantify the quality of reporting of future intervention adaptation studies. Finally, we provide a listing of examples of good reporting to assist the authors of future adaptation studies. However, our review has some limitations. The assessment of the quality of reporting of adaptation studies is challenging as it involves subjective judgement; The scoring of our ADAPT reporting checklist often required discussion between

the review team as to whether the reporting item was adequately met or not. Adaptation studies may be poorly indexed in databases, so there is a risk that potentially includable studies may have been overlooked by literature searches. This was evidenced by the fact that we identified 10 potentially relevant studies from reviewing the references of included studies; two of which were included in our final list of 25 studies [54, 55]. Non-English publication was pre-defined as an exclusion criterion of this review. However, we did not exclude any studies based on language. We acknowledge there was no formal patient or other stakeholder consultation as part of this scoping review.

Implications

Our findings have important implications. We confirm the importance of a systematic methodological approach to intervention adaptation and the need for high-quality reporting to enable healthcare professionals and programme planners to inform their implementation of adapted interventions. This can be particularly relevant in the transfer of healthcare interventions between LMICs and HICs where there are often fundamental differences in context and culture [57]. Without rigorous customization and adaptation, an intervention is likely not fit the context of the adapted intervention and its implementation is likely to be suboptimal. However, such a rigorous methodological approach requires adequate research resources, human capacity expertise, and funding, both of which are often scarce in a low-income country. As we have seen in our review, several studies have been based on global partnerships across academic institutions and research funding [57]. However, a sustained approach to intervention adaptation requires continued research investment and capacity building in a lower income context. It is also important to recognize the opportunities for bilateral knowledge transfer. With the challenge of exploding global healthcare costs, translating affordable and efficient approaches to interventional delivery from LMICs to HICs is likely to be increasingly important. It is key that future development of adaptation methods guidance considers the needs and priorities of the LMIC to inform high quality but also feasible research, and ultimately improve healthcare, in these regions [58, 59].

Conclusions

This scoping review presents a comprehensive and contemporary identification and synthesis of the international literature of complex intervention adaptation studies for people with LTCs. It provides a resource for researchers, policymakers, and practitioners working to adapt interventions to new contexts. Our review highlights two key developmental issues going forward: (i) the need for better conduct and reporting of all the stages of the adaptation process, including both the evaluation and implementation of an adapted intervention; and (ii) the future development of adaptation methods guidance considers the needs and priorities of LMICs.

Supplementary Material

Supplementary material is available at *Translational Behavioral Medicine* online.

Funding

This manuscript is based on PhD research work of JU which is funded by the University of Glasgow, Glasgow, UK.

Conflicts of interest statement. The authors declare that they have no conflicts of interest.

Human Rights

This article does not contain any studies with human participants performed by any of the authors.

Informed Consent

This study does not involve human participants and informed consent was therefore not required.

Welfare of Animals

This article does not contain any studies with animals performed by any of the authors.

Transparency Statements

Study registration: The protocol for the study was preregistered on the Open Science Framework website (https:// osf.io/vm7j4/). *Analytic plan pre-registration*: The analysis plan was not formally pre-registered.

Downloaded from https://academic.oup.com/tbm/advance-article/doi/10.1093/tbm/ibae031/7696116 by guest on 26 July 2024

Data Availability

All data relevant to the study are included in the article or uploaded as supplemental information. The presented research is a literature review of published data; there are no additional unpublished data. *Analytic code availability*: There is no analytic code associated with this study. Materials availability: Materials used to conduct the study are not publicly available.

References

- Marín-Maicas P, Corchón S, Ambrosio L *et al.* Living with long term conditions from the perspective of family caregivers. A scoping review and narrative synthesis. *Int J Environ Res Public Health* 2021;18:7294. https://doi.org/10.3390/ijerph18147294
- Hajat C, Stein E. The global burden of multiple chronic conditions: a narrative review. *Prev Med Rep* 2018;12:284–93. https://doi. org/10.1016/j.pmedr.2018.10.008
- Lall D, Engel N, Devadasan N *et al.* Models of care for chronic conditions in low/middle-income countries: a 'best fit' framework synthesis. *BMJ Global Health* 2018;3:e001077. https://doi. org/10.1136/bmjgh-2018-001077
- World Health Organization. Global status report on noncommunicable diseases 2014. World Health Organization; 2014. https://www.who.int/publications/i/item/9789241564854 (26 October 2023, date last accessed)
- GBD 2019 Diseases and Injuries Collaborators. Global burden of 369 diseases and injuries in 204 countries and territories, 1990– 2019: a systematic analysis for the Global Burden of Disease Study 2019. *Lancet (London, England)* 2020;396:1204–22. https://doi. org/10.1016/S0140-6736(20)30925-9
- Escoffery C, Lebow-Skelley E, Udelson H et al. A scoping study of frameworks for adapting public health evidence-based interventions. Transl Behav Med 2019;9:1–10. https://doi.org/10.1093/ tbm/ibx067
- Movsisyan A, Arnold L, Evans R *et al.* Adapting evidence-informed complex population health interventions for new contexts: a systematic review of guidance. *Implement Sci* 2019;14:105. https:// doi.org/10.1186/s13012-019-0956-5
- Stirman SW, Miller CJ, Toder K *et al.* Development of a framework and coding system for modifications and adaptations of evidence-based interventions. *Implement Sci* 2013;8:65. https://doi. org/10.1186/1748-5908-8-65
- Wiltsey Stirman S, Baumann AA, Miller CJ. The FRAME: an expanded framework for reporting adaptations and modifications to evidence-based interventions. *Implement Sci* 2019;14:58. https:// doi.org/10.1186/s13012-019-0898-y
- Beaglehole R, Epping-Jordan J, Patel V *et al.* Improving the prevention and management of chronic disease in low-income and middle-income countries: a priority for primary health care. *Lancet (London, England)* 2008;372:940–9. https://doi.org/10.1016/ S0140-6736(08)61404-X
- Ku GM, Kegels G. Adapting chronic care models for diabetes care delivery in low-and-middle-income countries: a review. World J Diabetes 2015;6:566–75. https://doi.org/10.4239/wjd.v6.i4.566
- Shah SC, Kayamba V, Peek RM Jr et al. Cancer control in low- and middle-income countries: is it time to consider screening? J Global Oncol 2019;5:1–8. https://doi.org/10.1200/JGO.18.00200
- Sundell K, Beelmann A, Hasson H et al. Novel programs, international adoptions, or contextual adaptations? Meta-analytical results from German and Swedish intervention research. J Clin Child Adolesc Psychol 2016;45:784–96. https://doi.org/10.1080/1 5374416.2015.1020540
- 14. Hasson H, Sundell K, Beelmann A et al. Novel programs, international adoptions, or contextual adaptations? Meta-analytical results from German and Swedish intervention research. BMC Health Serv Res 2014;14:1–2. https://doi.org/10.1186/1472-6963-14-S2-O32

- Peters MD, Godfrey CM, Khalil H *et al.* Guidance for conducting systematic scoping reviews. *Int J Evid Based Healthc* 2015;13:141– 6. https://doi.org/10.1097/XEB.0000000000000050
- Arksey H, O'Malley L. Scoping studies: towards a methodological framework. *Int J Soc Res Methodol* 2005;8:19–32. https://doi. org/10.1080/1364557032000119616
- 17. Levac D, Glegg SM, Sveistrup H *et al.* A knowledge translation intervention to enhance clinical application of a virtual reality system in stroke rehabilitation. *BMC Health Serv Res* 2016;16:557. https://doi.org/10.1186/s12913-016-1807-6
- Colquhoun HL, Levac D, O'Brien KK et al. Scoping reviews: time for clarity in definition, methods, and reporting. J Clin Epidemiol 2014;67:1291–4. http://dx.doi.org/10.1016/j.jclinepi.2014.03.013
- Tricco AC, Lillie E, Zarin W *et al.* PRISMA extension for scoping reviews (PRISMA-ScR): checklist and explanation. *Ann Intern Med* 2018;169:467–73. https://doi.org/10.7326/M18-0850
- 20. Movsisyan A, Arnold L, Copeland L et al. Adapting evidenceinformed population health interventions for new contexts: a scoping review of current practice. *Health Res Policy Syst* 2021;19:13. https://doi.org/10.1186/s12961-020-00668-9
- Payne RA, Mendonca SC, Elliott MN *et al*. Development and validation of the Cambridge Multimorbidity Score. *Can Med Assoc J journal de l'Association medicale canadienne* 2020;**192**:E107–14. https://doi.org/10.1503/cmaj.190757
- Barnett K, Mercer SW, Norbury M et al. Epidemiology of multimorbidity and implications for health care, research, and medical education: a cross-sectional study. Lancet (London, England) 2012;380:37–43. https://doi.org/10.1016/S0140-6736(12)60240-2
- Covidence. Covidence systematic review software, veritas health innovation. Melbourne, Australia. https://www.covidence.org/ (23 March 2023, date last accessed)
- 24. Craig P, Di Ruggiero E, Frohlich KL et al. Taking Account of Context in Population Health Intervention Research: Guidance for Producers, Users and Funders of Research.. National Institute for Health and Care Research; Wales, UK. 2018. https://www.facebook.com/watch?v=389331226876906, https://doi.org/10.3310/ cihr-nihr-01
- Moore G, Campbell M, Copeland L et al. Adapting interventions to new contexts-the ADAPT guidance. BMJ (Clin Res Ed.) 2021;374:n1679. https://doi.org/10.1136/bmj.n1679
- 26. Risendal B, Dwyer A, Seidel R *et al.* Adaptation of the chronic disease self-management program for cancer survivors: feasibility, acceptability, and lessons for implementation. J Cancer Educ 2014;29:762–71. https://doi.org/10.1007/s13187-014-0652-8
- 27. Risendal BC, Dwyer A, Seidel RW *et al.* Meeting the challenge of cancer survivorship in public health: results from the evaluation of the chronic disease self-management program for cancer survivors. *Front Public Health* 2015;2:214. https://doi.org/10.3389/ fpubh.2014.00214
- Cassel KD, Braun K, Ka'opua L et al. Samoan body and soul: adapting an evidence-based obesity and cancer prevention program. Qual Health Res 2014;24:1658–72. https://doi. org/10.1177/1049732314549021
- Chen EK, Reid MC, Parker SJ *et al.* Tailoring evidence-based interventions for new populations: a method for program adaptation through community engagement. *Eval Health Prof* 2013;36:73–92. https://doi.org/10.1177/0163278712442536
- 30. Parker SJ, Chen EK, Pillemer K et al. Participatory adaptation of an evidence-based, arthritis self-management program: making changes to improve program fit. Fam Community Health 2012;35:236–45. https://doi.org/10.1097/FCH.0b013e318250bd5f
- Magidson JF, Seitz-Brown CJ, Safren SA et al. Implementing behavioral activation and life-steps for depression and HIV medication adherence in a Community Health Center. Cogn Behav Pract 2014;21:386–403. https://doi.org/10.1016/j.cbpra.2013.10.002
- 32. Wechsberg WM, El-Bassel N, Carney T *et al.* Adapting an evidence-based HIV behavioral intervention for South African couples. *Subst Abuse Treat Prev Policy* 2015;10:6. https://doi.org/10.1186/s13011-015-0005-6

- 33. Williams AB, Wang H, Burgess J et al. Cultural adaptation of an evidence-based nursing intervention to improve medication adherence among people living with HIV/AIDS (PLWHA) in China. Int J Nurs Stud 2013;50:487–94. https://doi.org/10.1016/j. ijnurstu.2012.08.018
- 34. Atif N, Nazir H, Zafar S et al. Development of a psychological intervention to address anxiety during pregnancy in a low-income country. Front Psychiatry 2020;10:927. https://doi.org/10.3389/ fpsyt.2019.00927
- 35. O'Donnell A, Anderson P, Schmidt C et al. Tailoring an evidence-based clinical intervention and training package for the treatment and prevention of comorbid heavy drinking and depression in middleincome country settings: the development of the SCALA toolkit in Latin America. Glob Health Action 2022;15:2080344. https://doi. org/10.1080/16549716.2022.2080344
- 36. Bornheimer LA, Li Verdugo J, Holzworth J et al. Modifying a cognitive behavioral suicide prevention treatment for adults with schizophrenia spectrum disorders in community mental health. Psychiatry Res 2022;311:114505. https://doi.org/10.1016/j.psychres.2022.114505
- 37. Carver H, McCulloch P, Parkes T. How might the 'Icelandic model' for preventing substance use among young people be developed and adapted for use in Scotland? Utilising the consolidated framework for implementation research in a qualitative exploratory study. BMC Public Health 2021;21:1742. https://doi.org/10.1186/ s12889-021-11828-z
- 38. Cho D, Basen-Engquist K, Acquati C et al. Cultural adaptation of evidence-based lifestyle interventions for African American men with prostate cancer: a dyadic approach. Am J Mens Health 2020;14:1557988320945449. https://doi.org/10.1177/155 7988320945449
- Bertrand E, Naylor R, Laks J et al. Cognitive stimulation therapy for brazilian people with dementia: examination of implementation' issues and cultural adaptation. *Aging Mental Health* 2019;23:1400– 4. https://doi.org/10.1080/13607863.2018.1488944
- 40. Fort MP, Paniagua-Avila A, Beratarrechea A *et al.* Stakeholder engagement in the translation of a hypertension control program to Guatemala's public primary health care system: lessons learned, challenges, and opportunities. *Glob Heart* 2019;14:155–63. https:// doi.org/10.1016/j.gheart.2019.05.005
- 41. Greenberg J, Lin A, Zale EL *et al.* Development and early feasibility testing of a mind-body physical activity program for patients with heterogeneous chronic pain; The GetActive Study. *J Pain Res* 2019;12:3279–97. https://doi.org/10.2147/JPR.S222448
- Alvares Pereira G, Sousa I, Nunes MVS. Cultural adaptation of cognitive stimulation therapy (CST) for Portuguese people with dementia. *Clin Gerontol* 2022;45:891–902. https://doi.org/10.108 0/07317115.2020.1821857
- 43. Jans G, Lenzen S, Van Pottelbergh G et al. Self-management among community-dwelling people with chronic conditions: adapting evidence-based group programs using intervention mapping. Patient Educ Couns 2020;103:589–96. https://doi.org/10.1016/j. pec.2019.10.001
- 44. Muroff J, Robinson W, Chassler D et al. Use of a smartphone recovery tool for Latinos with co-occurring alcohol and other drug disorders and mental disorders. J Dual Diagn 2017;13:280–90. https://doi.org/10.1080/15504263.2017.1348649
- 45. Gorman JR, Drizin JH, Al-Ghadban FA et al. Adaptation and feasibility of a multimodal mindfulness-based intervention to promote sexual health in cancer survivorship. Transl Behav Med 2021;11:1885–95. https://doi.org/10.1093/tbm/ibab083
- 46. Kangovi S, Carter T, Charles D *et al.* Toward a scalable, patientcentered community health worker model: adapting the IMPaCT intervention for use in the outpatient setting. *Popul Health Manag* 2016;19:380–8. https://doi.org/10.1089/pop.2015.0157
- 47. Ojo T, Ryan N, Birkemeier J et al. Adapting a skills-based stroke prevention intervention for communities in Ghana: a qualitative study. Implement Sci Commun 2020;1:104. https://doi. org/10.1186/s43058-020-00084-8

- 48. Okoli RCB, Shedul G, Hirschhorn LR et al. Stakeholder perspectives to inform adaptation of a hypertension treatment program in primary healthcare centers in the Federal Capital Territory, Nigeria: a qualitative study. *Implement Sci Commun* 2021;2:97. https://doi. org/10.1186/s43058-021-00197-8
- 49. Olson J, Hadjiconstantinou M, Luff C et al. From the United Kingdom to Australia—adapting a web-based self-management education program to support the management of type 2 diabetes: Tutorial. J Med Internet Res 2022;24:e26339. https://doi. org/10.2196/26339
- 50. Tongsiri S, Levkoff S, Gallagher-Thompson D et al. Cultural adaptation of the reducing disability in Alzheimer's Disease (RDAD). Protocol for an intervention to reduce behavioral and psychological symptoms of dementia in Thailand. J Alzheimers Dis 2022;87:1603–14. https://doi.org/10.3233/JAD-215253
- 51. Hopkins SE, Orr E, Boyer BB et al. Culturally adapting an evidence-based intervention to promote a healthy diet and lifestyle for Yup'ik Alaska native communities. Int J Circumpolar Health 2023;82:2159888. https://doi.org/10.1080/22423982.2022.2159888
- 52. Tran HV, Nong HTT, Tran TTT *et al.* Adaptation of a problemsolving program (Friendship Bench) to treat common mental disorders among people living with HIV and AIDS and on methadone maintenance treatment in Vietnam: formative study. *JMIR Form Res* 2022;6:e37211. https://doi.org/10.2196/37211
- 53. Paniagua-Avila A, Fort MP, Glasgow RE et al. Evaluating a multicomponent program to improve hypertension control in

Guatemala: study protocol for an effectiveness-implementation cluster randomized trial. *Trials* 2020;21:509. https://doi.org/10. 1186/s13063-020-04345-8

- 54. Marinho V, Bertrand E, Naylor R et al. Cognitive stimulation therapy for people with dementia in Brazil (CST-Brasil): results from a single blind randomized controlled trial. Int J Geriatr Psychiatry 2021;36:286–93. https://doi.org/10.1002/gps.5421
- 55. Chen H, Levkoff S, Chuengsatiansup K et al. Implementation science in Thailand: design and methods of a geriatric mental health cluster-randomized trial. Psychiatr Serv (Washington, D.C.) 2022;73:83–91. https://doi.org/10.1176/appi.ps.202000028
- 56. Kirk MA, Moore JE, Wiltsey Stirman S et al. Towards a comprehensive model for understanding adaptations' impact: the model for adaptation design and impact (MADI). Implement Sci 2020;15:56. https://doi.org/10.1186/s13012-020-01021-y
- 57. Mounier-Jack S, Mayhew SH, Mays N. Integrated care: learning between high-income, and low- and middle-income country health systems. *Health Policy Plan* 2017;32:iv6–iv12. https://doi. org/10.1093/heapol/czx039
- Dye C, Reeder JC, Terry RF. Research for universal health coverage. *Sci Transl Med* 2013;5:199ed–13. https://doi.org/10.1126/ scitranslmed.3006971
- 59. Rosala-Hallas A, Bhangu A, Blazeby J *et al.* Global health trials methodological research agenda: results from a priority setting exercise. *Trials* 2018;19:48. https://doi.org/10.1186/s13063-018-2440-y